Author’s response to reviews

Title: Prevalence of diabetes and pre-diabetes in rural Tehri Garhwal, India: Influence of diagnostic method

Authors:

Pam Anderson (pamwanderson@gmail.com)
Nathan GRILLS (ngrills@gmail.com)
Rajesh SINGH (gcdwschc@gmail.com)
Rajkumari SINGH (rajkumari.singh39@gmail.com)
Roger EVANS (Roger.Evans@monash.edu)
Paramita SENGUPTA (drparamita2425@gmail.com)
Amanda THRIFT (amanda.thrift@monash.edu)

Version: 1 Date: 17 Feb 2019

Author’s response to reviews:

Written informed consent was obtained from each participant prior to collecting their data. This has now been clarified in the “Ethics approval and consent to participate” section of our Declarations.